Author’s response to reviews

Title: Sasang constitutional types for the risk prediction of metabolic syndrome: A 14-year longitudinal prospective cohort study

Authors:

Sunghee Lee (L.sunghee@gmail.com)
Chol Shin (chol-shin@korea.ac.kr)
Seung Ku Lee (leeseungku@gmail.com)
Jong Yeol Kim (ssmed@kiom.re.kr)
Namhan Cho (chnaha@ajou.ac.kr)

Version: 3 Date: 17 Aug 2017

Author’s response to reviews:

Editorial comments:

- Please include the email addresses of all manuscript authors on your Title Page.

Response: According to your suggestion, we included the email addresses of all authors on the title page.

- Thank you for uploading your figures into our online submission system. However, we would ask you to please remove the legends from these figures. The legends should instead be included as a list at the end of your manuscript document.

Response: According to your comment, we removed the legends from the figures and put the figure legend in the end of the manuscript. We do hope that this revised version is satisfactory for the publication of the Journal.